Author's response to reviews

Title: Thyroid abscess due to Aspergillosis: a case report

Authors:

Suemi Marui (suemimar@usp.br)
Ariella C.L. Pereira (ariellalima@yahoo.com.br)
Raquel M.A. Maia (uelxy@hotmamail.com)
Eduardo F Borba (eborba@lim17.ffm.usp.br)

Version: 6
Date: 11 August 2014

Author's response to reviews:

Dear Reviewer

As your suggestion, we added the Aspergillus features (highlighted in the text).

Thanks for your attention.

Sincerely,

Suemi Marui